Author's response to reviews

Title: Delayed Diagnosis of Intermittent Mesenteroaxial Volvulus of the Stomach by Computed Tomography: A Case Report

Authors:

Colin YL Woon (wolv23@gmail.com)
Alexander YF Chung (alexander.chung.y.f@sgh.com.sg)
Albert SC Low (gdrisc@sgh.com.sg)
Wai-Keong Wong (gsuwwk@sgh.com.sg)

Version: 2 Date: 30 December 2007

Author's response to reviews:

Under Authors' Contributions: Have amended that "ASCL provided analysis of the plain abdominal radiographs and CT images" instead of "magnetic resonance images".